Author's response to reviews

Title: Elevated soluble cellular adhesion molecules are associated with increased mortality in a prospective cohort of renal transplant recipients

Authors:

Grainne M Connolly (grainnemconnolly@aol.co.uk)
Ronan Cunningham (ronan.cunningham@northerntrust.hscni.net)
Peter T McNamee (peter.mcnamee@belfasttrust.hscni.net)
Ian S Young (i.young@qub.ac.uk)
Alexander P Maxwell (a.p.maxwell@qub.ac.uk)

Version: 4 Date: 18 May 2011

Author's response to reviews: see over
Dear Professor Norton

Manuscript: Elevated soluble cellular adhesion molecules are associated with increased mortality in a prospective cohort of renal transplant recipients

Authors: Grainne M. Connolly, Ronan Cunningham, Peter T. McNamee, Ian S. Young and Alexander P. Maxwell.

Thank you for accepting the above manuscript for publication in BMC Nephrology. The formatting changes requested by the Editorial office have been made to the uploaded manuscript.

Yours sincerely

Dr Grainne M Connolly
on behalf of the authors